Author's response to reviews

Title: Derivation and validation of a simple clinical bedside score (ATLAS) for Clostridium difficile infection which predicts response to therapy

Authors:

Mark A Miller (mmiller@lab.jgh.mcgill.ca)
Thomas Louie (thomas.louie@calgaryhealthregion.ca)
Kathleen Mullane (kmullane@medicinebsd.uchicago.edu)
Karl Weiss (weisscan@aol.com)
Arnold Lentnek (arnold.lentnek@earthlink.net)
Yoav Golan (ygolan@tuftsmedicalcenter.org)
Yin Kean (ymkean@yahoo.com)
Pam Sears (psears@optimerpharma.com)

Version: 4 Date: 5 February 2013

Author's response to reviews: see over